Author's response to reviews

Title: Key health promotion factors among male members of staff at a higher educational institution: a cross-sectional postal survey

Authors:
Alena Vasianovich (elenavasjanovich@yahoo.co.uk)
Edwin R van Teijlingen (van.teijlingen@abdn.ac.uk)
Garth Reid (g.reid@abdn.ac.uk)
Neil W Scott (n.w.scott@abdn.ac.uk)

Version: 5 Date: 24 January 2008

Author's response to reviews:

We have made all the formatting changes you requested from us. We have changed the section heading 'Strengths and Limitations' to Conclusions as requested but we had to (a) move that heading one paragraph down (to make sense to the reader) and (b) we had to add three words to the following sentence, the start of the paragraph, to make it read like a real conclusion.

Edwin van Teijlingen (on behalf of my fellow authors)